

# Onchocerca volvulus and epilepsy: A comprehensive review using the Bradford Hill criteria for causation

Robert Colebunders, Alfred Njamnshi, Sonia Menon, Charles Newton, An Hotterbeekx, Pierre-Marie Preux, Adrian Hopkins, Michel Vaillant, Joseph Nelson Siewe Fodjo

# ▶ To cite this version:

Robert Colebunders, Alfred Njamnshi, Sonia Menon, Charles Newton, An Hotterbeekx, et al.. Onchocerca volvulus and epilepsy: A comprehensive review using the Bradford Hill criteria for causation. PLoS Neglected Tropical Diseases, 2021, 15 (1), pp.e0008965. 10.1371/journal.pntd.0008965 . hal-03103704

# HAL Id: hal-03103704 https://unilim.hal.science/hal-03103704v1

Submitted on 5 Nov 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

REVIEW

# *Onchocerca volvulus* and epilepsy: A comprehensive review using the Bradford Hill criteria for causation

Robert Colebunders <sup>1</sup>\*, Alfred K. Njamnshi<sup>2,3,4</sup>, Sonia Menon<sup>1</sup>, Charles R. Newton<sup>5</sup>, An Hotterbeekx <sup>1</sup>, Pierre-Marie Preux<sup>6</sup>, Adrian Hopkins<sup>7</sup>, Michel Vaillant <sup>8</sup>, Joseph Nelson Siewe Fodjo<sup>1</sup>

 Global Health Institute, University of Antwerp, Antwerp, Belgium, 2 Neurology Department, Yaoundé Central Hospital, Yaoundé, Cameroon, 3 Faculty of Medicine and Biomedical Sciences, University of Yaoundé I, Yaoundé Cameroon, 4 Brain Research Africa Initiative (BRAIN), Yaoundé, Cameroon,
Department of Psychiatry, University of Oxford, Oxford, United Kingdom, 6 Institute of Epidemiology and Tropical Neurology, INSERM UMR1094, University of Limoges, Limoges, France, 7 Neglected and Disabling Diseases of Poverty Consultant, Kent, United Kingdom, 8 Competence Center in Methodology and Statistics, Luxembourg Institute of Health, Strassen, Luxembourg

\* robert.colebunders@uantwerpen.be

# Abstract

# Background

The possibility that onchocerciasis may cause epilepsy has been suggested for a long time, but thus far, an etiological link has not been universally accepted. The objective of this review is to critically appraise the relationship between *Onchocerca volvulus* and epilepsy and subsequently apply the Bradford Hill criteria to further evaluate the likelihood of a causal association.

# Methods

PubMed and gray literature published until September 15, 2020, were searched and findings from original research were synthesized. Adherence to the 9 Bradford Hill criteria in the context of onchocerciasis and epilepsy was determined to assess whether the criteria are met to strengthen the evidence base for a causal link between infection with *O. volvulus* and epilepsy, including the nodding syndrome.

# Results

Onchocerciasis as a risk factor for epilepsy meets the following Bradford Hill criteria for causality: strength of the association, consistency, temporality, and biological gradient. There is weaker evidence supporting causality based on the specificity, plausibility, coherence, and analogy criteria. There is little experimental evidence. Considering the Bradford Hill criteria, available data suggest that under certain conditions (high microfilarial load, timing of infection, and perhaps genetic predisposition), onchocerciasis is likely to cause epilepsy including nodding and Nakalanga syndromes.



# G OPEN ACCESS

**Citation:** Colebunders R, Njamnshi AK, Menon S, Newton CR, Hotterbeekx A, Preux P-M, et al. (2021) *Onchocerca volvulus* and epilepsy: A comprehensive review using the Bradford Hill criteria for causation. PLoS Negl Trop Dis 15(1): e0008965. https://doi.org/10.1371/journal. pntd.0008965

**Editor:** Sasisekhar Bennuru, National Institutes of Allergy and Infectious Diseases, NIH, UNITED STATES

Published: January 7, 2021

**Copyright:** © 2021 Colebunders et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** RC is funded by the European Research Council (ERC grant 671055). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

## Conclusion

Applying the Bradford Hill criteria suggests consistent epidemiological evidence that *O. volvulus* infection is a trigger of epilepsy. However, the pathophysiological mechanisms responsible for seizure induction still need to be elucidated.

# Author summary

There is growing epidemiological evidence that an infection with *Onchocerca volvulus* (the parasite causing onchocerciasis or river blindness) can trigger epilepsy, including nodding and Nakalanga syndromes. We studied the association between onchocerciasis and epilepsy using previously proposed criteria for causality and found strong arguments in favor of a causal relationship. The risk for children to develop epilepsy seems to be determined by the *O. volvulus* microfilarial load. However, the mechanism by which the *O. volvulus* parasite is able to trigger seizures remains to be elucidated. This form of epilepsy, called onchocerciasis-associated epilepsy (OAE), is an important public health problem in onchocerciasis-endemic regions with nonexistent or suboptimal onchocerciasis elimination programs. OAE is often characterized by sudden seizure onset in previously healthy children, between 3 to 18 years, without any other obvious cause of epilepsy. On the basis of the findings presented in this paper, it appears that onchocerciasis control may be able to prevent OAE. This further highlights the importance to eliminate onchocerciasis, particularly in areas with a high burden of epilepsy.

# Introduction

The possibility that onchocerciasis may cause epilepsy has been suggested since the pioneering work of Casis-Sacre in 1938 [1–5]; however, an etiological link is still pending universal acceptance. Hitherto, *Onchocerca volvulus* microfilariae, the small first-stage larval progeny produced by the adult female worms which are responsible for the disease manifestations of onchocerciasis (river blindness), have not been found in the brain parenchyma. However, microfilariae have been detected in cerebrospinal fluid in the past particularly after treatment with diethylcarbamazine [6,7]. *O. volvulus* is transmitted by tropical species of Simulium blackflies, which breed in fast-flowing waters, delimiting the distribution of onchocerciasis to areas in proximity to rivers in sub-Saharan Africa (SSA), some countries in Latin America, and Yemen. Epidemiological studies also suggested a potential link between nodding syndrome (NS), an epileptic encephalopathy, and onchocerciasis [8,9]. The pathogenesis of NS was recently discussed in an excellent review paper [10]. However, no systematic review that considers all the available original research data concerning the association between epilepsy including NS and onchocerciasis has so far been performed.

In 1965, Bradford Hill proposed 9 criteria for evaluating traditional epidemiologic data to "pass from an observed association to a verdict of causation," namely: strength of association, consistency, specificity, temporality, biological gradient, plausibility, coherence, experimental evidence, and analogy [11]. Building on the 9 criteria, other researchers have added excluding confounding factors and bias [12]. Since then, the Bradford Hill criteria have been frequently used in epidemiological studies to demonstrate causal inference, for example, Zika virus and microcephaly [13], sugar-sweetened beverages and coronary heart disease [14], and chrysotile asbestos and mesothelioma [15].

In this paper, we conduct a comprehensive review of published and unpublished literature and use Bradford Hill criteria to gauge the causality of association between infection with *O*. *volvulus* and epilepsy, including NS. We also highlight the criteria for which evidence is weaker or absent and identify research needs.

# Material and methods

We searched for articles indexed in PubMed until September 15, 2020. Search terms used included "epilepsy," "nodding syndrome/seizures," "Nakalanga syndrome/features," and "onchocerciasis" (see <u>S1 Table</u>). A manual search to identify relevant published articles as well as unpublished gray literature such as postgraduate research theses/dissertations was also performed. All original research reporting relevant quantitative findings were included in our review, irrespective of the study design. Data were extracted from eligible studies and organized in electronic spreadsheets.

# Causal association assessment

The possibility of a causal relationship between onchocerciasis and epilepsy (including nodding and Nakalanga syndromes [16]) was investigated by applying each of the 9 Bradford Hill criteria adapted to the case of onchocerciasis and epilepsy. We also discussed potential confounding factors and bias in the retrieved documentation (Table 1).

# Results

We retrieved 142 studies from PubMed (Fig 1). After screening the abstracts, 66 studies were eligible for inclusion in our review. Six additional findings from manual searches were included as well (S2 Table).

# Strength of the association

The strength of association has been examined in case-control studies, 2 meta-analysis, and 2 cohort studies [17,18]. In case-control studies, *O. volvulus* antibodies [8,9,19], skin snip positivity [9,19], high skin microfilarial load [3,19], onchocerciasis-associated skin disease [19,20], and presence of nodules [21,22] were observed more frequently in persons with epilepsy (PWE) than in controls.

Druet-Cabanac and colleagues reported no statistical association between O. volvulus infection and epilepsy in their meta-analysis of 9 papers searched until 2002 (odds ratio (OR): 1.21; 95% confidence interval (CI 0.99 to 1.47; p = 0.06) [23]. Kaiser and colleagues reported an OR of 2.5 with a 95% CI of 1.6 to 3.9 for an association between epilepsy and onchocerciasis in their meta-analysis of 11 case-control studies published by 2012, but after controlling for age, sex, and place of residence, the OR became nonsignificant (1.3; 95% CI 0.9 to 1.8) [24]. Taking into account these meta-analyses [23,24], one may conclude there is no association between onchocerciasis and epilepsy. However, 2 of the 9 studies included in the meta-analysis by Druet-Cabanac and colleagues are problematic with regard to the investigation of the association between O. volvulus infection and epilepsy. One study was performed in a village that was already hypo-endemic following more than 14 years of vector control activities by the Onchocerciasis Control Programme in West Africa (OCP) [25]. The other study by Ovuga and colleagues considered participants with retarded growth as non-epileptic controls [2]. However, persons with retarded growth are not appropriate controls, as growth retardation itself might be onchocerciasis-related [26]. If these 2 studies were to be excluded from the meta-analysis, a relative risk of 1.30 (95% CI 1.04 to 1.62; p = 0.02) would be obtained [27].

Criteria	Bradford Hill criteria [11]	Adaptation to <i>O. volvulus</i> infection and epilepsy, including NS and Nakalanga
1. Strength of association	The larger the association, the more likely that it is causal	The strength of the association between onchocerciasis and epilepsy/NS in studies carefully conducted to adjust for relevant confounders such as ivermectin use. An odds ratio >3.0 was considered as a strong association
2. Consistency	Consistent findings observed by different persons in different places with different samples strengthens the likelihood of an effect	The association between onchocerciasis and epilepsy/NS is replicated by different researchers in different countries/localities with different samples and where the net effect points toward the same direction
3. Specificity	Specificity is considered if an exposure is causing a very specific disease in a specific population at a specific site and with no other likely explanation. The more specific an association between an exposure and an effect, the bigger the probability of a causal relationship	Causation of epilepsy/NS can be considered if <i>O. volvulus</i> infection is causing the very specific disease manifestations in a specific population at a specific site with no other likely explanation, and if elimination of onchocerciasis eliminates incident cases of epilepsy/NS
4. Temporality	The effect has to occur after the cause	Epilepsy/NS occurs only following a documented infection with <i>O. volvulus</i>
5. Biological gradient	Greater exposure should generally lead to greater incidence of the effect	Greater prevalence (endemicity) and/or infection intensity of onchocerciasis should lead to greater incidence/severity of epilepsy/NS
6. <u>Plausibility</u>	A plausible biological mechanism between cause and effect is helpful	A plausible biological mechanism explaining how <i>O. volvulus</i> infection may cause epilepsy/NS
7. Coherence	Coherence between epidemiological and laboratory findings increases the likelihood of causality. Similar to biological plausibility, cause and effect should be in line with the knowledge available within the scientific community	Coherence between epidemiological and parasitological findings and the natural history of onchocerciasis and epilepsy/NS
8. Experiment	Experimental evidence from laboratory studies or randomized clinical trials	Experimental evidence (e.g., from animal models) showing that O. volvulus infection can lead to epilepsy/NS. Randomized clinical trials showing that treatment of onchocerciasis decreases the incidence of epilepsy/NS
9. Analogy	Causality is supported by analogy if there are similar associations or causal relationships in other areas of relevance; weakest form of evidence of causality.	A similar pathophysiological mechanism is involved as in other onchocerciasis-associated conditions such as skin and eye disease

Table 1. Bradford Hill criteria for assessing causation [11] and their adaptation to the case of onchocerciasis and epilepsy.

https://doi.org/10.1371/journal.pntd.0008965.t001

The OR reported in the meta-analysis by Kaiser and colleagues also needs to be interpreted with caution as case-control studies in hypo-endemic areas should be excluded from analysis because in such areas, the cause of epilepsy will be unrelated to onchocerciasis in the majority of PWE. In hyper-endemic areas that have not been exposed to ivermectin, it may be difficult to demonstrate an association between skin snip positivity and epilepsy because in such communities, almost everyone is infected. However, in such hyper-endemic settings, the intensity of O. volvulus infection (assessed via the microfilarial load) is often significantly higher among PWE compared to controls as demonstrated in the study by Boussinesq and colleagues [3]. Therefore, in the absence of past ivermectin treatment, a better parameter to compare PWE and controls would either be the microfilariae density or the number of nodules, both of which are proxies for the intensity of the onchocercal infection. A matched case-control study in an ivermectin-naïve population in an onchocerciasis-endemic region in Cameroon [3], and another in the Logo health zone in Ituri, Democratic Republic of Congo (DRC) [19], showed a higher microfilariae density among PWE compared to controls: 288 versus 141 microfilariae/ skin snip, P < 0.0001 in the Mbam valley (Cameroon) and 31.8 versus 2.7 microfilariae/skin snip, P < 0.001 in Draju, Logo health zone in the DRC.

In case of many years of community-directed treatment with ivermectin (CDTI) in a hyper-endemic area, the interpretation of results may be difficult because past ivermectin intake between cases and controls may have been different. Considering case-control studies conducted in meso- and hyper-endemic villages without previous exposure to ivermectin, a significant association between onchocerciasis and epilepsy/NS with an OR >3 was reported



#### Fig 1. Selection of eligible studies.

https://doi.org/10.1371/journal.pntd.0008965.g001

in almost all studies (Tables 2 and 3). The strongest association was observed in a study that included only NS cases (Table 2) [9]. This was to be expected because NS is a specific clinical entity reported to occur only in onchocerciasis foci, whereas other forms of epilepsy occur everywhere and can result from etiologies unrelated to *O. volvulus*.

In the Mbam valley of Cameroon, children aged 5 to 10 years from 25 villages had been tested for the presence and density of microfilariae in their skin snips in 1991 to 1993; our

Table 2.	Case-control studies inv	stigating the association	between onchocerciasis	and nodding syndrome
----------	--------------------------	---------------------------	------------------------	----------------------

Author, year of study, study site	Pre-control onchocerciasis endemicity level	Number of years of onchocerciasis control	Onchocerciasis endemicity level at the time of the study	OR <sup>#</sup> (95% CI) for presence of mf in cases vs controls
Foltz et al, 2009, Kitgum, Uganda [8]	Meso	0*	Meso	2.11 (0.86–5.2)
Tumwine et al, 2001, Amadi, South Sudan [9]	Hyper	Interrupted CDTI	Hyper	29.0 (3.5–237.7)
Tumwine et al, 2001, Lui, South Sudan [9]	Hyper	Interrupted CDTI	Hyper	9.3 (2.7–32.6)
Tumwine et al, 2002, Lui, South Sudan [9]	Hyper	Interrupted CDTI	Hyper	15.4 (1.6–148.8)
CDC, 2011, Maridi, South Sudan [28]	Hyper	Interrupted CDTI	Hyper	9.3 (1.9–52.3)
CDC, 2011, Witto, South Sudan [28]	Hyper	NA	Meso/hyper	1.0 (0.2–6.2)

#### <sup>#</sup>OR, unadjusted odds ratio.

Meso, meso-endemic (35% < mf prevalence < 60%); Hyper, hyper-endemic (mf prevalence >60%).

 $^{\ast}33\%$  of cases and 25% of controls had been treated with ivermectin.

CDC, Centers for Disease Control and Prevention; CDTI, community-directed treatment with ivermectin; mf, microfilariae; NA, not available.

https://doi.org/10.1371/journal.pntd.0008965.t002

Author, year of study, study site	Pre-control onchocerciasis endemicity level	Number of years of onchocerciasis control	Onchocerciasis endemicity level at the time of the study	OR <sup>#</sup> (95% CI) for presence of mf in cases vs controls
Boussinesq et al, 1991, Mbam, Cameroon [3]	Hyper (Pmf >69%)	0	Hyper (Pmf >69%)	4.18 (0.46-38.3)*
Newell et al, 1994, Bururi Province, Buyengero & Burambi, Burundi [29]	Meso-hyper	0	Meso/hyper	2.49 (1.38-4.5)
Mandro et al, 2015, Draju, Logo health zone, Ituri Province, DRC [19]	Meso	0	Meso	3.58 (1.68-7.63)
Kohler, 2000, Sanaga maritime, Cameroon [30]	Hyper	1	Hyper (Pmf >80%)	3.76 (1.31–10.74)
Kipp et al, 1993, Kabarole, Uganda [31]	1 hyper, 1 hypo	2	1 hyper, 1 hypo	7.31 (3.19–16.73)
Burfeind et al, 2014, Kasangulu, Bas-Congo Province, DRC [32]	Meso-hyper?	2	Meso-hyper?	4.51 (1.75–11.65)
Kaiser et al, 1994, Kabarole, Uganda [33]	7 hyper, 6 meso/hypo	3	7 hyper, 6 meso/hypo	1.67 (0.61-4.57)
Mandro et al, 2015, Rassia, Rethy health zone, Ituri Province, DRC [19]	Meso	3	Meso	2.13 (0.83–5.4)
Druet-Cabanac et al, 1996, Ouham and Ouham-Pende, Central African Republic [34]	Meso-hyper	5	Meso/hyper	1.17 (0.82–1.68)
König et al, 2005, Mahenge, Tanzania [35]	Meso-hyper	8	Meso	4.36 (2.63-7.24)
Gbenou, 1995, Agbogbome, Dassa-Zoumé, Benin [36]	Meso-hyper	8 (1 CDTI, 7 VC)	Meso (Pmf >47%)	2.85 (0.87–9.38)
Mandro et al, 2015, Salambongo, Wanieruklula, Tshopo Province, DRC [19]	Hyper	13	Hyper	1.89 (0.84–4.28)
Colebunders et al, 2014, Titule, Bas Uélé Province, DRC [20]	Meso	14	Meso	1.30 (0.32–5.30)

Table 3. Case-control studies in meso- and hyper-endemic areas on the onchocerciasis-epilepsy relationship (ranked according to increasing duration of onchocerciasis control.

<sup>#</sup>OR, unadjusted odds ratio.

Meso, meso-endemic (35% < Pmf  $<\!60\%$ ); Hyper, hyper-endemic (Pmf  $>\!60\%$ ).

\*144 study participants: Only 1/72 case and 4/72 controls were skin snip negative.

CDTI, community-directed treatment with ivermectin; DRC, Democratic Republic of Congo; NA, not available; Pmf, prevalence of microfiladermia; VC, vector control.

https://doi.org/10.1371/journal.pntd.0008965.t003

team revisited the previously surveyed villages and traced most of these children in 2017 (25 years later) [17]. Individual microfilariae densities during the initial parasitological surveys were strongly associated with the development of epilepsy later in life, with an adjusted incidence ratio of 28.5 (95% CI = 3.84 to 211.27) for developing epilepsy when childhood microfilariae densities exceeded 200 microfilariae/skin snip compared to individuals without detectable microfilaridermia in their skins during their early years [17]. A second cohort study conducted in other parts of Cameroon found similar results, further highlighting the role of *O. volvulus* in increasing the risk for epilepsy [18].

All studies showing an association between onchocerciasis and epilepsy need to be carefully examined for possible confounders. However, no consistent confounding factor has been reported that can explain the high prevalence of epilepsy observed in many onchocerciasis endemic areas across Africa. In Uganda, war, the internal displaced person (IDP) camps, and malnutrition could have played a role but these factors cannot be reported in most other onchocerciasis-endemic regions. Neurocysticercosis could be a confounder in the DRC, Tanzania, and Cameroon but not in Maridi (South Sudan) where there are no pigs for cultural reasons. In Maridi, we now have very strong arguments suggesting that the OAE/NS epidemic started after the building of the Maridi dam, a blackly breeding site [37].

Given the high OR observed in case-control studies without previous ivermectin exposure in onchocerciasis meso- and hyper-endemic villages, and the very high incidence ratio observed in the cohort studies, we should consider the first Bradford Hill criterion to be met.

## Consistency

Causation is more likely if the results from various research studies are consistent. This criterion requires examining all studies included in the review to see whether similar conclusions have been drawn. As early as 1938, Casis-Sacre described a syndrome characterized by epileptic seizures, stunted growth, and mental retardation in patients with onchocerciasis in the Chiapas and Oaxaca foci of Mexico [1]. Positive associations between onchocerciasis and epilepsy have been demonstrated in *O. volvulus*-endemic areas throughout West [38–40], Central [19,41–43], and East Africa [9,33,44,45]. Although the epileptogenic role of neurocysticercosis was suggested as an explanation in some areas [46], case-control studies did not reveal a significant difference in the prevalence of *Taenia solium* antibodies between cases and controls in those areas [47]. Other authors, analyzing case-control studies in Uganda and South Sudan where NS incidence had increased in the decade preceding their study, concluded that there was a consistent, yet "enigmatic" association with onchocerciasis (detected by skin snip or serological diagnosis) [48].

In Tanzania, the prevalence of epilepsy was 3.5% in 2 rural villages located close to a blackfly-infested river compared to 1.5% in 2 suburban villages (P < 0.001) [49] despite 20 years of community-directed treatment with ivermectin (CDTI). The use of a rapid diagnostic test, applied to children aged 6 to 10 years, detected a higher seroprevalence of IgG4 antibodies against the Ov16 antigen of *O. volvulus* in these rural villages (42.6%) compared to the suburban villages (4.7%), P < 0.001, indicating greater exposure to onchocerciasis in the villages with a higher prevalence of epilepsy [49].

A multisite study of active convulsive epilepsy was conducted in 5 sub-Saharan Africa Health and Demographic Surveillance System (HDSS) sites, including onchocerciasis-endemic sites in 3 countries (Ghana, Tanzania, and Uganda) [50,51]. In 2 of them (Ghana and Tanzania), IgG4 seropositivity to Ov16 antigen by ELISA (sensitivity 90%; specificity 98%) was statistically and positively associated with epilepsy in children (aged <18 years) and adults (18 years and older). Among children <18 years, the population attributable fraction of epilepsy due to O. volvulus seropositivity was significant in Ghana (Kintampo HDSS = 0.14, 95% CI = 0.01 to 0.25) and Tanzania (Ifakara HDSS = 0.09, 95% CI = 0.01 to 0.17), but not in Uganda (Iganga-Mayuge HDSS = 0.05, 95% CI = 0 to 0.10) [50]. The Iganga-Maguye HDSS in Uganda comprises of districts which were no longer endemic for onchocerciasis, in contrast with Kintampo HDSS in Ghana, and Ifakara HDSS in Tanzania where onchocerciasis was still endemic at the time of the study. However, we must note that in this large study, simultaneous exposure to O. volvulus and other parasites including Toxoplasma gondii increased the risks for convulsive epilepsy [52]. In case-control studies in South Sudan and Uganda, both O volvulus and Mansonella perstans antibodies were more prevalent in NS cases than in controls [8,9] but this was not observed in a case-control study in an onchocerciasis endemic area in the Bas Uélé Province in the DRC [20]. Moreover, in a recent cohort study in Cameroon, childhood infection with *M. perstans* was not found to be a risk factor for developing epilepsy later in life, in contrast to infection with O. volvulus [18].

A high prevalence and incidence of epilepsy was observed particularly in onchocerciasisendemic regions where *O. volvulus* transmission was either poorly or not at all controlled [3,9,22,41,43,53–55]. In a study of 23 villages in a rural area of Cameroon, the closer a village was to the Mbam river (a known habitat for simuliid species such as those in the *Simulium*  *squamosum* group, which are competent vectors of *O. volvulus*), the higher the prevalence of epilepsy (Pearson's correlation coefficient r = 0.465, P = 0.026) [3]. In Titule, in the Bas-Uélé Province of the DRC, proximity of households to blackfly-infested rivers was also a risk factor for epilepsy, with increasing distance being negatively associated with the prevalence of epilepsy (OR = 0.63, 95% CI = 0.45 to 0.91, P < 0.05) [56]. In the Central African Republic, a study conducted in 1996 reported a positive and significant correlation (r = 0.73; P < 0.001) between onchocerciasis endemicity levels determined by community microfilarial levels (hypoendemic, mesoendemic, and hyperendemic) and epilepsy prevalence (respectively, 0.5%, 0.8%, and 2.5%), and a negative and significant correlation (r = -0.34; P < 0.03) between epilepsy prevalence and mean distance from the nearest river of the villages in each endemicity level (respectively, 9.5, 5.9, and 2.4 Km) [57]. A positive correlation between epilepsy and onchocerciasis, albeit being weaker, was also observed in the Imo river basin in Nigeria: (r = 0.38; P < 0.043) [39,58].

The low number of reports about high epilepsy prevalence in onchocerciasis-endemic regions in West Africa is most likely explained by the success of OCP that was started in 1974 to 1975 [59]. In 1981, a high prevalence of epilepsy in West Africa was reported from an onchocerciasis-endemic region in Liberia (4.9%) [38], a country that was not under the umbrella of the OCP. In contrast, in 1990, in an area of Burkina Faso that was previously hyper-endemic but had become hypo-endemic under OCP [59], the prevalence of onchocerciasis in PWE was not significantly different from that in persons without epilepsy (P = 0.67), and the epilepsy prevalence was 1.5% (95% CI = 0.9% to 2.3%), was similar to the 1.1% recorded in a non-endemic area [25]. A recent meta-analysis of epilepsy prevalence studies performed in West Africa showed that before and during the early years of implementing onchocerciasis control in West Africa, high onchocerciasis endemicity was associated with a high prevalence of epilepsy and that subsequent control efforts significantly reduced the prevalence of epilepsy. Higher pre-control endemicity and a shorter duration of onchocerciasis control were both associated with increased epilepsy prevalence (P < 0.001) [40]. By applying the OAE criteria on the epidemiological and clinical data obtained from 2 epilepsy studies in Ivory Coast [60,61], we found that >70% of PWE in the study villages reported features of OAE. It is therefore plausible that these sites were OAE hotspots at the time of the study, when onchocerciasis transmission was still high [40].

The notion that OAE may have been a possible cause of excess mortality in the West African countries covered by the OCP is suggested by the study of Walker and colleagues [62] in which 295,909 persons with onchocerciasis recorded in the OCP cohort database were followup for 25 years: The relative risk of mortality increased with increasing skin microfilariae load and this risk was significantly higher in those aged <20 years compared to those  $\geq$ 20 years old [62]. Onchocercal eye disease cannot explain these excess deaths in younger people, as it is more frequent in adulthood [63]. Given that seizure onset between the ages of 3 to 18 years is characteristic for both NS [48] and OAE [64], we surmise that the high mortality observed among those <20 years old could be due to premature mortality in PWE living in these areas [65,66].

While a statistically significant association between onchocerciasis and epilepsy has been demonstrated in areas where onchocerciasis is meso- or hyper-endemic [3,5,9,54], some case-control studies in settings with lower endemicity have not reported such an association [25,34,67]. This could be due to the fact that epilepsy is a chronic condition which persists even after treating or controlling onchocerciasis. Moreover, epilepsy onset is often accompanied by dramatic life changes such as decreased exposure to the river and blackflies (because of the risk of drowning in the river) and increased ivermectin use as was observed in the DRC [41,68]. Consequently, these factors may influence results regarding *O. volvulus* 

microfilaridermia status and microfilarial density in the case-control studies which enroll PWE with a certain duration of epilepsy [56].

In the case-control study in the Bas-Congo Province in the DRC, an association between onchocerciasis and epilepsy was found (Table 2), but when stunting was included as a correcting covariate, the relationship between onchocerciasis and epilepsy was no longer significant [32]. The results of this study are, however, difficult to interpret; indeed only 22 controls were included, there was no information about previous ivermectin use, and there is a problem of multicollinearity between epilepsy and stunting as both conditions have been reported to be associated with onchocerciasis [64].

In summary, the association between epilepsy and onchocerciasis has been observed in a variety of settings and populations, spanning from conflict zones with displaced populations and episodes of food insecurity to stable settings with adequate food supply. Study results not showing an association between onchocerciasis and epilepsy/NS can be explained by low onchocerciasis endemicity, prior onchocerciasis control interventions including ivermectin treatment, not taking into account participants' microfilarial load during the data analysis, and/or recruiting controls with onchocerciasis-associated morbidity. As 100% of studies not meeting the latter descriptions support the association, this criterion is met.

# Specificity

According to Bradford Hill, causation is more likely if there is a specific outcome related to a specific exposure. Onchocerciasis indeed seems to be associated with a specific type of epilepsy with an onset of seizures of unknown etiology in previously healthy children between the ages of 3 to 18 years (with a peak onset between the ages 8 to 12 years), with some children presenting with nodding seizures and Nakalanga features [44,64,69–71]. This age of epilepsy onset between 3 and 18 years, although not exclusive to OAE, is very characteristic and is not frequently observed in non-onchocerciasis–endemic regions in Africa [50]. In the latter, most people develop epilepsy before the age of 5 years due to perinatal causes or epilepsy of genetic origin [50] (Fig 2).

The pattern of age at OAE onset is also different from that of seizures caused by neurocysticercosis. Neurocysticercosis is characterized by a later onset of epilepsy with more persons experiencing their first seizure after the age of 20 years [72] (Fig 3).

The data on epilepsy prevalence in onchocerciasis-endemic regions need to be interpreted with caution in the absence of additional tests such as neuroimaging to differentiate OAE from epilepsy due to neurocysticercosis and other causes. However, in a recent epilepsy survey of 17,652 persons living in Maridi county, an onchocerciasis-hyper-endemic area in South Sudan where there are no pigs, an epilepsy prevalence of 4.4% was observed with more than 85% of PWE meeting the criteria of OAE ( $\geq 2$  seizures without any obvious cause, starting between the ages of 3 to 18 years in previously healthy persons who had resided for at least 3 years in an onchocerciasis meso- or hyper-endemic area) [54]. This high percentage of epilepsy in Maridi, most likely caused by OAE, is similar to the estimated high population attributable fraction of OAE (91.7%, 95% CI 56.7 to 98.4; P = 0.0021) in the cohort study in Cameroon [17]. Additionally, a second cohort study ruled out infection with *Loa loa* and *M. perstans* as the cause of the epilepsy burden observed in Cameroonian villages [18].

Causation is also more likely if a specific population, at a specific site, presents with the condition and when no other likely explanation exists beyond the suspected trigger. In other words, the specificity criterion relates to whether the type of epilepsy in onchocerciasisendemic regions could be attributed to anything other than onchocerciasis. NS and Nakalanga syndrome have only been reported in onchocerciasis-endemic areas, particularly in villages



Fig 2. The ages of village residents at epilepsy onset (A) in non-onchocerciasis–endemic regions in Kilifi, Kenya and Agincourt, South Africa [50] (B) in onchocerciasis–endemic communities in Mbam, Cameroon [43], Ituri, DRC [53], and Mahenge, Tanzania [49].

https://doi.org/10.1371/journal.pntd.0008965.g002



**Fig 3.** The ages at seizure onset in persons with neurocysticercosis (blue) and those in onchocerciasis-endemic areas (orange). Adapted from findings reported by Monteiro and colleagues [72] and by Lenaerts and colleagues [53].

https://doi.org/10.1371/journal.pntd.0008965.g003





#### https://doi.org/10.1371/journal.pntd.0008965.g004

where onchocerciasis is hyper-endemic and together with a high prevalence of other forms of epilepsy [64]. A suspected case of NS was reportedly seen in India [73], but the clinical presentation was substantially marred by preexisting neurological conditions, thus making the diagnosis of NS very unlikely. While other infectious, nutritional, or environmental causes have been suggested for the NS [74,75], the most significant association was consistently found with *O. volvulus* infection [76].

According to Bradford Hill, causation is more likely if altering the cause modifies the disease outcome. This criterion also seems to be met as onchocerciasis elimination efforts have been shown to decrease the incidence of epilepsy [77], including NS [43,78]. Indeed, in northern Uganda (Kitgum, Pader, and Lamwo districts), no new cases of NS have been reported after the implementation of biannual treatment with ivermectin coupled with ground-based larviciding of the rivers in which the vectors breed since 2013 [78]. This is in contrast with South Sudan, where CDTI has frequently been interrupted and the NS epidemic is ongoing [37,54]. In several villages in the Mbam valley in Cameroon, an age shift toward older PWE has been observed; in fact, the peak age-specific prevalence of epilepsy shifted from the 5 to 19 years age group to 20 to 39 years age group after 19 years of CDTI [42,43] (Fig 4).

This shift to older ages suggests a decreased incidence of OAE in the 5 to 19 years age group. Improvement of perinatal and child care in Bilomo cannot explain this increase in age because since 1998, the prevalence of epilepsy in the children aged below 10 years was already very low. A similar age shift was observed in 2017 in northern Uganda [78]. These findings, however, should be confirmed in a prospective, population-based study.

In 1994, a study in Kabarole, an onchocerciasis-endemic focus in western Uganda documented a prevalence of epilepsy of 3.0%, and reported cases of NS [79]. Following the elimination of onchocerciasis in this area since 2004, the same villages were revisited in 2018 and no new cases of NS were found among persons who were never exposed to *O. volvulus*. We equally observed a significant drop in the overall epilepsy prevalence (from 3.0% to 1.2%) and incidence after onchocerciasis elimination (from 418 to 73 new cases per 100,000 personyears), with the greatest impact on the 10 to 19 years age group which is usually the most affected by OAE [79]. A thorough overview of possible changes in the common risk factors for epilepsy in the Kabarole villages (perinatal brain insult, cerebral malaria, neurocysticercosis, and genetics) could not explain the substantial reduction in the epilepsy burden, leaving the elimination of onchocerciasis as the most plausible explanation [79].

It is noteworthy that only a minority of children with O. volvulus infection develop epilepsy; more often they develop skin manifestations such as itching and dermatitis [80]. Long-term exposure to high microfilarial loads is necessary to become blind, with the incidence of blindness being significantly and positively associated with increasing microfilarial load [81]. The threshold microfilarial density for developing epilepsy is probably lower than that for blindness (>20 microfilariae/skin snip has been suggested, above which there is severe ocular morbidity [82]), although it likely still needs to be sufficiently high [17,83]. Hence, the different clinical presentations of OAE (NS, Nakalanga syndrome, and other forms of epilepsy) may be related to the age at which children become infected with O. volvulus and their microfilarial density. In a recent study in South Sudan, persons with NS were found to have more disabilities and to have higher microfilarial densities than persons with other forms of OAE [84]. In areas where Onchocerca ochengi is endemic among cattle, humans when bitten by O. ochengiinfected blackflies may develop some immunity toward O. volvulus [85,86], and therefore, in such areas, children may develop lower O. volvulus microfilarial loads and consequently less NS and Nakalanga which are characterized by high microfilarial densities. The fact that other zoonotic filarial infections may play a similar protective role is not excluded and may influence the clinical spectrum of OAE. Genetic diversity of the O. volvulus, human genetics, immunological, and nutritional risk factors as well as other coinfections could also be incriminated. Although the available evidence supports that O. volvulus infection may cause epilepsy, more studies are warranted to better investigate in detail other confounding epilepsy etiologies in onchocerciasis foci. We do not consider this criterion to be met.

# Temporality

Until recently, all studies investigating the association between onchocerciasis and epilepsy were cross-sectional and therefore unable to demonstrate the temporal nature of causality. A first retrospective cohort study conducted in the Mbam valley of Cameroon showed that children in onchocerciasis-endemic areas first acquire *O. volvulus* infection and later develop epilepsy, particularly those with a high microfilarial load, in a dose-response relationship [17]. A second retrospective cohort study was performed in the Lékié division, an area of Cameroon where *L. loa* and onchocerciasis are co-endemic, which confirmed the temporal relationship between onchocerciasis and epilepsy [18].

At the population level, an NS epidemic started around the year 2000 in the districts of Kitgum, Pader, and Lamwo in northern Uganda, in an onchocerciasis-endemic region where ivermectin was not distributed. By 2008 after initiation of CDTI, the number of new NS cases started to decrease [78]. The reason why an NS epidemic started in northern Uganda is most likely multifactorial. Contributing factors to consider are the close proximity of IDP camps to the blackfly breeding sites, no access to ivermectin, and the poor nutritional status of the children during the war, making them susceptible to more severe *O. volvulus* infection as well as other health conditions. Another event that may have played a role is the theft in 1986 of about 300,000 cattle from the Acholi people in northern Uganda [87]. The presence of a large number of cattle in a village may decrease *O. volvulus* transmission and onchocerciasis-associated morbidity. Indeed before 1986, cattle were grazing the high grass along the rivers exposing blackflies to the sun and reducing their lifespan. Moreover, humans will be less bitten by blackflies because blackflies are an alternative food source. Finally, the hypothesized crossimmunity mechanisms that occur when humans are bitten by *O. ochengi*-infected blackflies may confer some protection against *O. volvulus* [85,86].

In 1948, DJ Lewis, a medical entomologist, described Mvolo in Western Equatoria in South Sudan as a place with extremely intense *Simulium* spp. biting and high infection prevalence in the flies (up to 10% of flies with L3 larvae in the heads) [88]. He described Mvolo as having only a police post with very few people residing in the area. Today Mvolo is a rural city, where a rapid assessment revealed that about 50% of the families have at least 1 child with epilepsy [89]. It is hypothesized that despite the risk for onchocerciasis, many people settled in Mvolo because of the fertile grounds located close to the Naam river and because of the great fishing opportunities. In a recent door-to-door survey in Maridi county in South Sudan, 692 (4.7%) of 4,619 permanent household members were found to have epilepsy compared to 82 (2.7%) of 3,033 immigrant household members; among the latter group, the prevalence of epilepsy increased with increasing duration of residence in the endemic village [54]. We consider this Bradford Hill criterion to be met.

# **Biological gradient (dose response)**

The severity of the cause should be positively correlated with the severity of the effect. This criterion appears to be met. Indeed, the risk for an individual of developing epilepsy increases in a dose-related manner with increasing microfilarial load [17,18] (as does the relative risk of mortality in children aged <20 years [62]) in onchocerciasis-endemic (or previously endemic) areas. At the community level, a higher prevalence of epilepsy has been observed in villages in Cameroon with a higher mean community microfilarial load [3]. Based on a meta-analysis of 8 population-based epilepsy surveys performed in 7 different countries, Pion and colleagues [5] calculated that, on average, the prevalence of epilepsy increased by 0.4% for each 10% increase in onchocerciasis prevalence. A higher microfilarial density was found to be associated with NS, a more severe form of OAE [84].

Onchocerciasis elimination efforts, which lead to reductions in microfilarial prevalence and load, have been shown to decrease the incidence of epilepsy/NS [78,79]. Recent studies in the DRC showed that high seizure frequency was associated with increasing microfilarial density [71] and increasing levels of urinary N-acetyltyramine-O, $\beta$ -glucuronide (NATOG), a biomarker for onchocerciasis [90]. Furthermore, a randomized trial in the DRC, comparing a single versus a multiple dose of ivermectin treatment regimen in *O. volvulus*-infected persons with epilepsy treated with phenobarbital showed that the multiple dose ivermectin regimen was associated with a reduced frequency of seizures [91]. The PWE included in this trial were all initially treated with phenobarbital despite having different seizure types, which may have affected the overall study outcomes. Given the complexity of the trial, these results need to be interpreted with caution. However, this trial certainly shows that treatment with ivermectin is not a possible confounder that could explain the high prevalence of epilepsy in onchocerciasis-endemic regions. We consider this Bradford Hill criterion is met.

# Plausibility

This criterion is met if a hypothesized effect makes sense in the context of current biological knowledge. There are 2 plausible biological mechanisms that could explain how *O. volvulus* infection is able to cause epilepsy/NS. The first mechanism is that *O. volvulus* infection induces the production of neurotoxic autoantibodies. Recently, it was shown that leiomodin-1 antibodies were more often present, and at higher levels, in sera of Ugandan and South Sudanese children with NS compared to controls [92]. These antibodies were also present in the

cerebrospinal fluid of children with NS. They were found to be neurotoxic in vitro and crossreact with *O. volvulus*-specific proteins (a phenomenon known as molecular mimicry) [92]. The leiomodin-1 protein is expressed in specific neuronal populations in the brain, including cortical neurons in the CA3 region of the hippocampus, and the Purkinje cells in the cerebellum [93]. However, leiomodin-1 is a member of the actin filament nucleator family that is highly enriched in smooth muscle-containing tissues such as the artery wall and the gastrointestinal tract [94]. If NS was an autoimmune disease triggered by leiomodin-1 antibodies, one would expect symptoms related to vasculitis, which is not reported in NS and was not observed in 2 postmortem studies [95,96]. So far, the Leiomodin-1 hypothesis has not been confirmed and therefore cannot be used as an argument for plausibility.

A recently reported case-control study including 30 South Sudanese persons with NS and a similar number of healthy participants from the same geographical region revealed autoimmune antibodies to 3 extracellular peptides of ionotropic glutamate receptors in NS patients: AMPA-GluR3B peptide antibodies (86%), NMDA-NR1 peptide antibodies (77%), and NMDA-NR2 peptide antibodies (87%) [97]. However, these antibodies were also observed in some controls at a lower concentration and were previously found in many more patients with other types of epilepsy and neurological conditions [98]. Therefore, the connection between these antibodies and NS is not clear. The same research group showed that in South Sudan, NS was associated with both protective HLA haplotypes: HLA-B\*42:01, C\*17:01, DRB1\*03:02, DQB1\*04:02, and DQA1\*04:01, and the susceptible motif: Ala24, Glu63, and Phe67, in the HLA-B peptide-binding groove. The authors therefore suggested that different HLA molecules may explain why under similar environmental conditions, only some children within the same families, tribes, and districts would develop NS, while others do not [99].

A second mechanism that needs to be explored is that *O. volvulus* microfilariae themselves or parasite-derived factors can cross the blood–brain barrier and cause neuronal damage, either due to their direct neurotoxicity or by provoking a secondary inflammatory response. Before the introduction of CDTI, the presence of *O. volvulus* microfilariae in the cerebrospinal fluid was reported in heavily infected patients, particularly after treatment with diethylcarbamazine [6,7,100]. However, recent studies on cerebrospinal fluid of patients with NS and epilepsy in onchocerciasis-endemic regions have failed to identify *O. volvulus* microfilariae or DNA [35,76,101,102].

A recently published postmortem study of 5 persons from northern Uganda who died of NS between 2014 and 2017 suggested that NS is a tauopathy and a neurodegenerative disease [95]. However, in another postmortem study performed in 2017 to 2018, among 5 persons with NS and 4 persons with another form of OAE who died in the same region, no evidence of a tauopathy was revealed [96]. Therefore, tau deposits are most likely the consequence and not the cause of the disease. In fact, tau pathology can be induced by seizures themselves [103] as well as by seizure-associated phenomena including hypoxia [104] and repeated head injuries [105]. We do not consider that this Bradford Hill criterion is met.

# Coherence

Causation is more likely if clinical observations are supported by and in agreement with the natural history of the disease. After becoming infected with *O. volvulus* and without ivermectin treatment, microfilarial loads will increase over time because exposure to infection and reinfection continues [63], and the adult parasite is very long-lived. It is mainly between the ages of 3 to 18 years, during the period in which children develop OAE, that the rate of increase in microfilarial load is more pronounced [81]. We surmise that around a median age of 8 years, some children, already heavily infected at a young age, may develop NS associated with severe

cognitive impairment [48,106], while others, less heavily infected, and/or infected at a later age, may develop other forms of epilepsy associated with less disability around the age of 11 years [106]. Similar to other infectious diseases, not all persons develop the disease after even a major exposure to a certain pathogen. It is possible that individuals who do not develop epilepsy once a certain threshold of microfilarial density is reached, for genetic and/or immuno-logical reasons, will never develop epilepsy later.

Furthermore, between 3 and 18 years, a large neuronal remodeling occurs in the brain of children and adolescents, with the formation of new neuronal connections and disappearance of others. It is possible that during this period, brain cells are particularly vulnerable.

We do not consider that this Bradford Hill criterion is met.

## **Experimental evidence**

*O. volvulus* infections only occur in humans, and this is a challenge for the development of a suitable animal model. Regarding evidence potentially provided by clinical trials, it is not ethically possible to conduct a prospective cohort study comparing the incidence of epilepsy in *O. volvulus*-infected individuals treated or not treated with ivermectin. However, a prospective study is planned in South Sudan to demonstrate whether implementation of ivermectin twice yearly (biannual CDTI) will decrease the incidence of OAE in a highly endemic area [107].

We do not consider that this Bradford Hill criterion is met.

## Analogy

As stated above, *O. volvulus* infection only causes disease in humans. *O. ochengi*, a parasite of cattle and the closest relative of *O. volvulus* [108], is transmitted by the same blackfly vector species (*Simulium damnosum sensu lato*) but is not known to cause epilepsy. Other species of *Onchocerca* can cause disease in animals and sometimes in humans (e.g., *O. lupi*), but thus far, these infections have not been associated with epilepsy [109]. Only one of the other filarial infections (*L. loa*) is able to cause encephalopathy, but this is related to microfilariae dying in the peripheral circulation mostly as a result of microfilaricidal treatment of heavily infected individuals. A similar phenomenon is unlikely in onchocerciasis because, in contrast with loasis, *O. volvulus* microfilariae are not usually present in blood [80,110].

Hookworm infections, like onchocerciasis, are acquired during the childhood, when the exposure is high. They both seem to affect the childhood development and some cognitive functions [111]. Also, for pinworms (*Enterobius vermicularis*), there is an old known effect on the behavior of heavily infected children [112].

*O. volvulus* may cause epilepsy in a similar way as it causes blindness. Lesions of the anterior part of the eye are directly caused by the dying microfilariae, and probably also by an inflammatory reaction toward the endosymbiont *Wolbachia* released by dying microfilariae, as both the filariae and endobacteria contribute to the pathogenesis of onchocerciasis [113]. The pathophysiological mechanism that causes the retinal lesions observed in *O. volvulus*-infected persons remains, however, poorly understood. The incidence of these ocular lesions increases and the average age of onset decreases as the intensity of transmission and infection in the community rises [114]. Microfilariae have been observed in the retina [115] but it has also been suggested that retinal lesions are caused by an autoimmune process whereby *O. volvulus* antibodies react with retinal proteins [116]. The *O. volvulus* antigen Ov39 is cross-reactive with the retinal antigen hr44 and induces ocular inflammation in rats after immunization [117]. A similar mechanism of molecular mimicry may play a role in NS as discussed with leiomodin-1 and *O. volvulus* proteins [92]. In retinal onchocerciasis, the inflammatory process can continue and visual impairment persists despite ivermectin treatment. In OAE, it still

needs to be determined whether early antiepileptic and antifilarial treatment may alter or even reverse the observed progressive evolution toward an encephalopathy in some individuals. We do not consider that this Bradford Hill criterion is met.

# Discussion

This paper reviews current evidence for the causal association between epilepsy/NS and onchocerciasis. Onchocerciasis as a risk factor for epilepsy meets the following Bradford Hill criteria for causality: strength of the association, consistency, temporality, and biological gradient. There is also weaker evidence for causality concerning the criteria: specificity, plausibility, coherence, and analogy. Until now, there is little experimental evidence, and this is identified as an important research gap that needs further in vivo, in vitro, and epidemiological studies such as those discussed here [92,107,118,119]. However, consideration of the Bradford Hill criteria suggests that there is consistent evidence that *O. volvulus*, in the presence of a sufficiently high microfilarial load, is able to trigger epilepsy including NS and Nakalanga features. Onchocerciasis is associated with a large number of clinical manifestations that include not only skin and ocular lesions, but also different types of epilepsy, intellectual disabilities, stunted growth, facial, thoracic, and spinal abnormalities, and delayed or absence of sexual development. *O. volvulus*-infected individuals may present with one or a combination of these manifestations [64].

Our extensive literature review pointed out the confounding role of CDTI during previous studies investigating this subject, because frequent ivermectin use has the ability to mask an association between onchocerciasis and epilepsy [19]. While debated in the epidemiological community [120], the Bradford Hill criteria are still widely accepted as useful guidelines for investigating causality in epidemiological studies [121]. As the world of epidemiological research is evolving, our criteria for determining causal inference must similarly evolve to reflect the multidisciplinary research needed to establish a causal association. Further research is needed to identify cofactors, such as parasitic coinfections, nutritional and (epi)genetic factors that may increase the risk for O. volvulus infected-children to develop epilepsy. The OAE definition mentioned severally in this paper is a useful tool to estimate the burden of disease caused by onchocerciasis besides skin and eye disease. Although of little diagnostic value, OAE criteria offer a convenient public health approach to identifying hotspots where onchocerciasis elimination efforts need to be strengthened [122]. However, to identify the pathophysiological mechanism of OAE, the preferred study population should be persons with NS since it is a specific, typical, and well-characterized onchocerciasis-related epileptic syndrome [123]. Based on studies with other nematode infections, we can hypothesize that certain O. volvulus excretory/ secretory molecules may have an effect on the human brain and its functions. Different cofactors may be involved in different onchocerciasis-endemic regions, explaining differences in the epidemiology and clinical spectrum of OAE. Screening for neurotoxic autoantibodies or O. volvulus-secreted proteins in blood and cerebrospinal fluid samples of persons with new onset epilepsy in onchocerciasis-endemic regions and controls, as well as additional postmortem studies of persons who died during an earlier phase of the disease, may be the way to identify the pathophysiological mechanism(s) by which O. volvulus may cause epilepsy/NS. Ultimately, we may need an experimental animal model to confirm this (these) mechanism(s).

# Conclusions

Applying the Bradford Hill criteria, there appears to be consistent epidemiological evidence that *O. volvulus* infection can cause epilepsy, including NS and Nakalanga features. The pathophysiological mechanism by which *O. volvulus* affects the central nervous system is unclear

and represents a research gap in this domain. Although understanding the underlying biological mechanism of OAE is important, it should not stop us from taking appropriate public health action [124]. Bradford Hill himself, stated "what is biologically plausible depends upon the biological knowledge of the day" [11]. Therefore, while this enigma is being solved and onchocerciasis eventually becomes known as "river epilepsy" rather than "river blindness," public health efforts must be stepped up to effectively control and eliminate onchocerciasis and to manage epilepsy in the affected and suffering populations.

# **Supporting information**

S1 Table. Search strategy for papers reporting epilepsy, nodding/Nakalanga syndrome, and onchocerciasis in PubMed (search date: September 15, 2020). (PDF)

**S2** Table. Literature retrieved via PubMed and manual searchesTab. (XLSX)

# Acknowledgments

We thank Robin Ryder for suggesting the idea of using the Bradford Hill criteria to evaluate the evidence discussed in this paper. We thank Maria-Gloria Basáñez for reading over the manuscript and for her useful comments.

## References

- 1. Casis S. El Sindrome Epileptico y sus reaciones con Onchocercosis. Boletin de Salubridad e Higiene 1938; 1.
- Ovuga E, Kipp W, Mungherera M, Kasoro S. Epilepsy and retarded growth in a hyperendemic focus of onchocerciasis in rural western Uganda. East Afr Med J 1992; 69(10):554–6. PMID: 1473507.
- Boussinesq M, Pion SD, Demanga N, Kamgno J. Relationship between onchocerciasis and epilepsy: a matched case-control study in the Mbam Valley, Republic of Cameroon. Trans R Soc Trop Med Hyg 2002; 96(5):537–41. https://doi.org/10.1016/s0035-9203(02)90433-5 PMID: 12474484.
- 4. Jilec-Aall L. Epilepsy and onchocerciasis: Pioneering research of Mexican physicians vindicated Investigacion en Salud. 2004; 6 (1):22–7.
- Pion SD, Kaiser C, Boutros-Toni F, Cournil A, Taylor MM, Meredith SE, et al. Epilepsy in onchocerciasis endemic areas: systematic review and meta-analysis of population-based surveys. PLoS Negl Trop Dis 2009; 3(6):e461. https://doi.org/10.1371/journal.pntd.0000461 PMID: 19529767; PubMed Central PMCID: PMC2691484.
- Duke BO, Vincelette J, Moore PJ. Microfilariae in the cerebrospinal fluid, and neurological complications, during treatment of onchocerciasis with diethylcarbamazine. Tropenmed Parasitol 1976; 27 (2):123–32. PMID: 941247.
- 7. Mazotti. Presencia de microfilarias de Onchocerca volvulus en el liquido cefalorraquideo de enfermos tratados con hetrazan. Rev Inst Salubr Enferm Trop. 1959; 19:1–5.
- Foltz JL, Makumbi I, Sejvar JJ, Malimbo M, Ndyomugyenyi R, Atai-Omoruto AD, et al. An Epidemiologic Investigation of Potential Risk Factors for Nodding Syndrome in Kitgum District, Uganda PLoS One 2013; 8(6):e66419. https://doi.org/10.1371/journal.pone.0066419 PMID: 23823012; PubMed Central PMCID: PMC3688914.
- Tumwine JK, Vandemaele K, Chungong S, Richer M, Anker M, Ayana Y, et al. Clinical and epidemiologic characteristics of nodding syndrome in Mundri County, southern Sudan. Afr Health Sci 2012; 12 (3):242–8. https://doi.org/10.4314/ahs.v12i3.1 PMID: 23382736; PubMed Central PMCID: PMC3557692.
- Johnson TP, Sejvar J, Nutman TB, Nath A. The Pathogenesis of Nodding Syndrome. Annu Rev Pathol 2020; 15:395–417. https://doi.org/10.1146/annurev-pathmechdis-012419-032748 PMID: 31977293
- 11. Bradford Hill AB. The environment and disease: association or causation? Proc R Soc Med 1965; 58.
- 12. Fedak KM, Bernal A, Capshaw ZA, Gross S. Applying the Bradford Hill criteria in the 21st century: how data integration has changed causal inference in molecular epidemiology. Emerg Themes Epidemiol

2015; 12:14. https://doi.org/10.1186/s12982-015-0037-4 PMID: 26425136; PubMed Central PMCID: PMC4589117.

- Frank C, Faber M, Stark K. Causal or not: applying the Bradford Hill aspects of evidence to the association between Zika virus and microcephaly. EMBO Mol Med. 2016; 8(4):305–7. https://doi.org/10. 15252/emmm.201506058 PMID: 26976611; PubMed Central PMCID: PMC4818755.
- Huffman MD. Association or causation of sugar-sweetened beverages and coronary heart disease: recalling Sir Austin Bradford Hill. Circulation 2012; 125(14):1718–20. https://doi.org/10.1161/ CIRCULATIONAHA.112.097634 PMID: 22412071.
- Lemen RA. Chrysotile asbestos as a cause of mesothelioma: application of the Hill causation model. Int J Occup Environ Health 2004; 10(2):233–9. https://doi.org/10.1179/oeh.2004.10.2.233 PMID: 15281385.
- Foger K, Gora-Stahlberg G, Sejvar J, Ovuga E, Jilek-Aall L, Schmutzhard E, et al. Nakalanga Syndrome: Clinical Characteristics, Potential Causes, and Its Relationship with Recently Described Nodding Syndrome. PLoS Negl Trop Dis 2017; 11(2):e0005201. https://doi.org/10.1371/journal.pntd. 0005201 PMID: 28182652; PubMed Central PMCID: PMC5300103.
- Chesnais CB, Nana-Djeunga HC, Njamnshi AK, Lenou-Nanga CG, Boulle C, Bissek AZ, et al. The temporal relationship between onchocerciasis and epilepsy: a population-based cohort study. Lancet Infect Dis 2018; 18(11):1278–86. https://doi.org/10.1016/S1473-3099(18)30425-0 PMID: 30268645.
- Chesnais CB, Bizet C, Campillo JT, Njamnshi WY, Bopda J, Nwane P, et al. A Second Population-Based Cohort Study in Cameroon Confirms the Temporal Relationship Between Onchocerciasis and Epilepsy. Open Forum Infect Dis. 2020; 7(6):ofaa206. <u>https://doi.org/10.1093/ofid/ofaa206</u> PMID: 32587878; PubMed Central PMCID: PMC7304933.
- Mandro M, Suykerbuyk P, Tepage F, Rossy D, Ngave F, Hasan MN, et al. Onchocerca volvulus as a risk factor for developing epilepsy in onchocerciasis endemic regions in the Democratic Republic of Congo: a case control study. Infect Dis Poverty 2018; 7(1):79. https://doi.org/10.1186/s40249-018-0465-9 PMID: 30293530; PubMed Central PMCID: PMC6174565.
- Colebunders R, Mandro M, Mokili JL, Mucinya G, Mambandu G, Pfarr K, et al. Risk factors for epilepsy in Bas-Uele Province, Democratic Republic of the Congo: a case-control study. Int J Infect Dis 2016; 49:1–8. https://doi.org/10.1016/j.ijid.2016.05.018 PMID: 27210267; PubMed Central PMCID: PMC4973807.
- Pion SD, Boussinesq M. Significant association between epilepsy and presence of onchocercal nodules: case-control study in Cameroon. Am J Trop Med Hyg 2012; 86(3):557; author reply 8. https://doi. org/10.4269/ajtmh.2012.11-0603a PMID: 22403337; PubMed Central PMCID: PMC3284382.
- Prischich F, De Rinaldis M, Bruno F, Egeo G, Santori C, Zappaterreno A, et al. High prevalence of epilepsy in a village in the Littoral Province of Cameroon. Epilepsy Res 2008; 82(2–3):200–10. https://doi.org/10.1016/j.eplepsyres.2008.09.004 PMID: 18976884.
- Druet-Cabanac M, Boussinesq M, Dongmo L, Farnarier G, Bouteille B, Preux PM. Review of epidemiological studies searching for a relationship between onchocerciasis and epilepsy. Neuroepidemiology 2004; 23(3):144–9. https://doi.org/10.1159/000075958 PMID: 15084784.
- Kaiser C, Pion SD, Boussinesq M. Case-control studies on the relationship between onchocerciasis and epilepsy: systematic review and meta-analysis. PLoS Negl Trop Dis 2013; 7(3):e2147. https://doi. org/10.1371/journal.pntd.0002147 PMID: 23556028; PubMed Central PMCID: PMC3610636.
- 25. Kabore JK, Cabore JW, Melaku Z, Druet-Cabanac M, Preux PM. Epilepsy in a focus of onchocerciasis in Burkina Faso. Lancet 1996; 347(9004):836. https://doi.org/10.1016/s0140-6736(96)90917-4 PMID: 8622372.
- 26. Kipp W, Burnham G, Bamuhiiga J, Leichsenring M. The Nakalanga syndrome in Kabarole District, Western Uganda. Am J Trop Med Hyg 1996; 54(1):80–3. https://doi.org/10.4269/ajtmh.1996.54.80 PMID: 8651377.
- Siewe Fodjo JN, Tepage F, Colebunders R. Epidemiological evidence concerning the association between onchocerciasis and epilepsy. Int J Infect Dis 2019; 82:77–8. <u>https://doi.org/10.1016/j.ijid.</u> 2019.03.015 PMID: 30885705.
- Centers for Disease Control and Prevention. Nodding Syndrome—South Sudan, 2011. Morb Mortal Wkly Rep. 2012; 61(3). PMID: 22278159
- Newell ED, Vyungimana F, Bradley JE. Epilepsy, retarded growth and onchocerciasis, in two areas of different endemicity of onchocerciasis in Burundi. Trans R Soc Trop Med Hyg 1997; 91(5):525–7. https://doi.org/10.1016/s0035-9203(97)90009-2 PMID: 9463656.
- Kohler M. Onchocercoseund epilepsie Eine Fallkontrol studie in der landlichen Regiion Sanaga-Maritime in Kamerun. MD thesis: University Wurzburg, Germany, pp118 2009.

- Kipp W, Kasoro S, Burnham G. Onchocerciasis and epilepsy in Uganda. Lancet 1994; 343 (8890):183–4. https://doi.org/10.1016/s0140-6736(94)90980-6 PMID: 7904037.
- Burfeind KG, Kashama JK, Bora BK, Murchison CF, Ramos-Crawford AL, Nseka MT, et al. Baseline characterization of epilepsy in an onchocerciasis endemic area of the Democratic Republic of Congo. Brain Res Bull 2019; 145:45–52. https://doi.org/10.1016/j.brainresbull.2018.11.009 PMID: 30468846; PubMed Central PMCID: PMC6377286.
- 33. Kaiser C, Rubaale T, Tukesiga E, Kipp W, Kabagambe G, Ojony JO, et al. Association between onchocerciasis and epilepsy in the Itwara hyperendemic focus, West Uganda: controlling for time and intensity of exposure. Am J Trop Med Hyg 2011; 85(2):225–8. https://doi.org/10.4269/ajtmh.2011.10-0544 PMID: 21813839; PubMed Central PMCID: PMC3144817.
- Druet-Cabanac M, Preux PM, Bouteille B, Bernet-Bernady P, Dunand J, Hopkins A, et al. Onchocerciasis and epilepsy: a matched case-control study in the Central African Republic. Am J Epidemiol 1999; 149(6):565–70. https://doi.org/10.1093/oxfordjournals.aje.a009853 PMID: 10084246.
- 35. Konig R, Nassri A, Meindl M, Matuja W, Kidunda AR, Siegmund V, et al. The role of Onchocerca volvulus in the development of epilepsy in a rural area of Tanzania. Parasitology 2010; 137(10):1559–68. https://doi.org/10.1017/S0031182010000338 PMID: 20388236.
- 36. Gbenou HD. Contribution à létudede l'association onchocercose épileposiue. Résultats préliminaires d'une enqueteneuroépidémiologiqueà Agbogbpome, communede Paouignan, sous-préfecturede Dassa-zoumé, au Bénin. MD theis, Natinal University of Benin 126pp. 1995.
- Lakwo TL, Raimon S, Tionga M, Siewe Fodjo JN, Alinda P, Sebit WJ, et al. The Role of the Maridi Dam in Causing an Onchocerciasis-Associated Epilepsy Epidemic in Maridi, South Sudan: An Epidemiological, Sociological, and Entomological Study. Pathogens. 2020; 9(4). https://doi.org/10.3390/ pathogens9040315 PMID: 32344586; PubMed Central PMCID: PMC7238195.
- Gerrits C. A West African epilepsy focus. Lancet 1983; 1(8320):358. <u>https://doi.org/10.1016/s0140-6736(83)91663-x</u> PMID: 6130361.
- Dozie IN, Onwuliri CO, Nwoke BE, Chukwuocha UM, Chikwendu CI, Okoro I, et al. Onchocerciasis and epilepsy in parts of the Imo river basin, Nigeria: a preliminary report. Public Health 2006; 120 (5):448–50. https://doi.org/10.1016/j.puhe.2005.10.010 PMID: 16524608.
- 40. Siewe Fodjo JN RJ, Preux PM, Colebunders R. Meta-analysis of epilepsy prevalence in West Africa and its relationship with onchocerciasis endemicity and control. Int Health. 2020;March 6 pii: ihaa012. https://doi.org/10.1093/inthealth/ihaa012 PMID: 32141502
- Levick B, Laudisoit A, Tepage F, Ensoy-Musoro C, Mandro M, Bonareri Osoro C, et al. High prevalence of epilepsy in onchocerciasis endemic regions in the Democratic Republic of the Congo. PLoS Negl Trop Dis 2017; 11(7):e0005732. https://doi.org/10.1371/journal.pntd.0005732 PMID: 28708828; PubMed Central PMCID: PMC5529017.
- Boulle C, Njamnshi AK, Dema F, Mengnjo MK, Siewe Fodjo JN, Bissek AZ, et al. Impact of 19 years of mass drug administration with ivermectin on epilepsy burden in a hyperendemic onchocerciasis area in Cameroon. Parasit Vectors 2019; 12(1):114. https://doi.org/10.1186/s13071-019-3345-7 PMID: 30890155; PubMed Central PMCID: PMC6423875.
- 43. Siewe Fodjo JN, Tatah G, Tabah EN, Ngarka L, Nfor LN, Chokote SE, et al. Epidemiology of onchocerciasis-associated epilepsy in the Mbam and Sanaga river valleys of Cameroon: impact of more than 13 years of ivermectin. Infect Dis Poverty 2018; 7(1):114. https://doi.org/10.1186/s40249-018-0497-1 PMID: 30501640; PubMed Central PMCID: PMC6276171.
- 44. Bhwana D, Mmbando BP, Dekker MC, Mnacho M, Kakorozya A, Matuja W, et al. Clinical presentation of epilepsy in six villages in an onchocerciasis endemic area in Mahenge, Tanzania Epileptic Disord 2019; 21(5):425–35. https://doi.org/10.1684/epd.2019.1093 PMID: 31649005.
- Kaiser C, Kipp W, Asaba G, Mugisa C, Kabagambe G, Rating D, et al. The prevalence of epilepsy follows the distribution of onchocerciasis in a west Ugandan focus. Bull World Health Organ 1996; 74 (4):361–7. PMID: 8823957; PubMed Central PMCID: PMC2486881.
- 46. Katabarwa M, Lakwo T, Habumogisha P, Richards F, Eberhard M. Could neurocysticercosis be the cause of "onchocerciasis-associated" epileptic seizures? Am J Trop Med Hyg 2008; 78(3):400–1. PMID: 18337333.
- Kaiser C, Pion S, Preux PM, Kipp W, Dozie I, Boussinesq M. Onchocerciasis, cysticercosis, and epilepsy. Am J Trop Med Hyg 2008; 79(5):643–4; author reply 4–5. PMID: <u>18981494</u>.
- Dowell SF, Sejvar JJ, Riek L, Vandemaele KA, Lamunu M, Kuesel AC, et al. Nodding syndrome. Emerg Infect Dis 2013; 19(9):1374–84. https://doi.org/10.3201/eid1909.130401 PMID: 23965548; PubMed Central PMCID: PMC3810928.
- 49. Mmbando BP, Suykerbuyk P, Mnacho M, Kakorozya A, Matuja W, Hendy A, et al. High prevalence of epilepsy in two rural onchocerciasis endemic villages in the Mahenge area, Tanzania, after 20 years of

community directed treatment with ivermectin. Infect Dis Poverty 2018; 7(1):64. https://doi.org/10. 1186/s40249-018-0450-3 PMID: 29921319; PubMed Central PMCID: PMC6009039.

- Ngugi AK, Bottomley C, Kleinschmidt I, Wagner RG, Kakooza-Mwesige A, Ae-Ngibise K, et al. Prevalence of active convulsive epilepsy in sub-Saharan Africa and associated risk factors: cross-sectional and case-control studies. Lancet Neurol 2013; 12(3):253–63. https://doi.org/10.1016/S1474-4422(13) 70003-6 PMID: 23375964; PubMed Central PMCID: PMC3581814.
- Ae-Ngibise KA, Akpalu B, Ngugi A, Akpalu A, Agbokey F, Adjei P, et al. Prevalence and risk factors for Active Convulsive Epilepsy in Kintampo, Ghana Pan Afr Med J 2015; 21:29. https://doi.org/10.11604/ pamj.2015.21.29.6084 PMID: 26401223; PubMed Central PMCID: PMC4561141.
- Kamuyu G, Bottomley C, Mageto J, Lowe B, Wilkins PP, Noh JC, et al. Exposure to multiple parasites is associated with the prevalence of active convulsive epilepsy in sub-Saharan Africa. PLoS Negl Trop Dis 2014; 8(5):e2908. https://doi.org/10.1371/journal.pntd.0002908 PMID: 24875312; PubMed Central PMCID: PMC4038481.
- Lenaerts E, Mandro M, Mukendi D, Suykerbuyk P, Dolo H, Wonya'Rossi D, et al. High prevalence of epilepsy in onchocerciasis endemic health areas in Democratic Republic of the Congo. Infect Dis Poverty 2018; 7(1):68. https://doi.org/10.1186/s40249-018-0452-1 PMID: 30064504; PubMed Central PMCID: PMC6069757.
- Colebunders R, J YC, Olore PC, Puok K, Bhattacharyya S, Menon S, et al. High prevalence of onchocerciasis-associated epilepsy in villages in Maridi County, Republic of South Sudan: A communitybased survey. Seizure. 2018; 63:93–101. https://doi.org/10.1016/j.seizure.2018.11.004 PMID: 30468964; PubMed Central PMCID: PMC6291739.
- Kaiser C, Asaba G, Leichsenring M, Kabagambe G. High incidence of epilepsy related to onchocerciasis in West Uganda. Epilepsy Res 1998; 30(3):247–51. <u>https://doi.org/10.1016/s0920-1211(98)</u> 00007-2 PMID: 9657652.
- Colebunders R, Tepage F, Rood E, Mandro M, Abatih EN, Musinya G, et al. Prevalence of River Epilepsy in the Orientale Province in the Democratic Republic of the Congo. PLoS Negl Trop Dis 2016; 10 (5):e0004478. https://doi.org/10.1371/journal.pntd.0004478 PMID: 27139245; PubMed Central PMCID: PMC4854481.
- Druet-Cabanac MBI. Preux PM. Onchocerciasis and epilepsy in the North West Region of Central African Republic Afri J Neuro/Sci 1998; 17:37–8.
- Dozie I, Onwuliri C, Nwoke B. Onchocerciasis in Imo state, Nigeria (2): the prevalence, intensity and distribution in the upper Imo river basin. Int J Environ Health Res 2004; 14(5):359–69. <u>https://doi.org/</u> 10.1080/09603120400004048 PMID: 15385215.
- 59. Boatin B. The Onchocerciasis Control Programme in West Africa (OCP). Ann Trop Med Parasitol 2008; 102 Suppl 1:13–7. https://doi.org/10.1179/136485908X337427 PMID: 18718148.
- Kaudjhis PJR. Les agrégats de l'épilepsie de M'brou: approche électroclinique et étiologique. MD Thesis, Université Nationale de Côte d'Ivoire. 1995.
- **61.** Kouadjo Y. Génétique et épilepsie: à propos d'un foyer d'épilepsie observe dans un village lvoirien. MD Thesis, Université Nationale de Côte d'Ivoire. 1990.
- Walker M, Little MP, Wagner KS, Soumbey-Alley EW, Boatin BA, Basanez MG. Density-dependent mortality of the human host in onchocerciasis: relationships between microfilarial load and excess mortality. PLoS Negl Trop Dis 2012; 6(3):e1578. https://doi.org/10.1371/journal.pntd.0001578 PMID: 22479660; PubMed Central PMCID: PMC3313942.
- **63.** Renz A, Fuglsang H, Anderson J. Studies on the dynamics of transmission of onchocerciasis in a Sudan-savanna area of North Cameroon IV. The different exposure to Simulium bites and transmission of boys and girls and men and women, and the resulting manifestations of onchocerciasis. Ann Trop Med Parasitol 1987; 81(3):253–62. https://doi.org/10.1080/00034983.1987.11812118 PMID: 3662667.
- Colebunders R, Nelson Siewe FJ, Hotterbeekx A. Onchocerciasis-Associated Epilepsy, an Additional Reason for Strengthening Onchocerciasis Elimination Programs. Trends Parasitol 2018; 34(3):208– 16. https://doi.org/10.1016/j.pt.2017.11.009 PMID: 29288080.
- **65.** Kamgno J, Pion SD, Boussinesq M. Demographic impact of epilepsy in Africa: results of a 10-year cohort study in a rural area of Cameroon. Epilepsia 2003; 44(7):956–63. <u>https://doi.org/10.1046/j. 1528-1157.2003.59302.x PMID: 12823580</u>.
- 66. Kaiser C, Asaba G, Kasoro S, Rubaale T, Kabagambe G, Mbabazi M. Mortality from epilepsy in an onchocerciasis-endemic area in West Uganda. Trans R Soc Trop Med Hyg 2007; 101(1):48–55. https://doi.org/10.1016/j.trstmh.2006.06.003 PMID: 16905166.
- **67.** Farnarier G, Diop S, Coulibaly B, Arborio S, Dabo A, Diakite M, et al. [Onchocerciasis and epilepsy. Epidemiological survey in Mali]. Med Trop (Mars) 2000; 60(2):151–5. PMID: <u>11100441</u>.

- Mukendi D, Tepage F, Akonda I, Siewe JNF, Rotsaert A, Ndibmun CN, et al. High prevalence of epilepsy in an onchocerciasis endemic health zone in the Democratic Republic of the Congo, despite 14 years of community-directed treatment with ivermectin: A mixed-method assessment. Int J Infect Dis 2019; 79:187–94. https://doi.org/10.1016/j.ijid.2018.10.021 PMID: 30711145.
- Kaiser C, Rubaale T, Tukesiga E, Kipp W, Asaba G. Nodding syndrome, western Uganda, 1994. Am J Trop Med Hyg. 2015; 93(1):198–202. https://doi.org/10.4269/ajtmh.14-0838 PMID: 25918208; PubMed Central PMCID: PMC4497897.
- Siewe JFN, Ngarka L, Tatah G, Mengnjo MK, Nfor LN, Chokote ES, et al. Clinical presentations of onchocerciasis-associated epilepsy (OAE) in Cameroon. Epilepsy Behav 2018; 90:70–8. <a href="https://doi.org/10.1016/j.yebeh.2018.11.008">https://doi.org/10.1016/j.yebeh.2018.11.008</a> PMID: 30513438.
- 71. Siewe Fodjo JN, Mandro M, Mukendi D, Tepage F, Menon S, Nakato S, et al. Onchocerciasis-associated epilepsy in the Democratic Republic of Congo: Clinical description and relationship with microfilarial density. PLoS Negl Trop Dis 2019; 13(7):e0007300. https://doi.org/10.1371/journal.pntd.0007300 PMID: 31314757; PubMed Central PMCID: PMC6663032.
- 72. Monteiro L, Nunes B, Mendonca D, Lopes J. Spectrum of epilepsy in neurocysticercosis: a long-term follow-up of 143 patients. Acta Neurol Scand 1995; 92(1):33–40. <u>https://doi.org/10.1111/j.1600-0404.1995.tb00463.x PMID: 7572058</u>.
- 73. Sehgal R, Agarwal N, Gera R. Nodding syndrome- an Indian case. Afr Health Sci 2018; 18(3):837–41. https://doi.org/10.4314/ahs.v18i3.43 PMID: 30603018; PubMed Central PMCID: PMC6306992.
- 74. Spencer PS, Mazumder R, Palmer VS, Lasarev MR, Stadnik RC, King P, et al. Environmental, dietary and case-control study of Nodding Syndrome in Uganda: A post-measles brain disorder triggered by malnutrition? J Neurol Sci 2016; 369:191–203. https://doi.org/10.1016/j.jns.2016.08.023 PMID: 27653888.
- 75. Lagoro DK, Arony DA. Nodding syndrome (NS) and Onchocerca Volvulus (OV) in Northern Uganda. Pan Afr Med J 2017; 28:1. https://doi.org/10.11604/pamj.2017.28.1.13554 PMID: 29138647; PubMed Central PMCID: PMC5681003.
- 76. Spencer PS, Vandemaele K, Richer M, Palmer VS, Chungong S, Anker M, et al. Nodding syndrome in Mundri county, South Sudan: environmental, nutritional and infectious factors. Afr Health Sci 2013; 13 (2):183–204. <u>https://doi.org/10.4314/ahs.v13i2.2</u> PMID: <u>24235915</u>; PubMed Central PMCID: PMC3824482.
- 77. Siewe JNF, Ukaga CN, Nwazor EO, Nwoke MO, Nwokeji MC, Onuoha BC, et al. Low prevalence of epilepsy and onchocerciasis after more than 20 years of ivermectin treatment in the Imo River Basin in Nigeria. Infect Dis Poverty 2019; 8(1):8. https://doi.org/10.1186/s40249-019-0517-9 PMID: 30670093; PubMed Central PMCID: PMC6343278.
- 78. Gumisiriza N, Mubiru F, Siewe Fodjo JN, Mbonye Kayitale M, Hotterbeekx A, Idro R, et al. Prevalence and incidence of nodding syndrome and other forms of epilepsy in onchocerciasis-endemic areas in northern Uganda after the implementation of onchocerciasis control measures. Infect Dis Poverty 2020; 9(1):12. https://doi.org/10.1186/s40249-020-0628-3 PMID: 32114979; PubMed Central PMCID: PMC7050130.
- **79.** Gumisiriza N KC, Asaba G, Onen H, Mubiru F, Kisembo D, Siewe Fodjo JN, Colebunders R. Onchocerciasis elimination drastically reduces epilepsy prevalence and incidence in Kabarole, Western Uganda. Abstract, 4th African epilepsy conference, 22–24 August, Entebbe, Uganda 2019.
- 80. Burnham G. Onchocerciasis Lancet 1998; 351(9112):1341–6. https://doi.org/10.1016/S0140-6736 (97)12450-3 PMID: 9643811.
- Little MP, Breitling LP, Basanez MG, Alley ES, Boatin BA. Association between microfilarial load and excess mortality in onchocerciasis: an epidemiological study. Lancet 2004; 363(9420):1514–21. https://doi.org/10.1016/S0140-6736(04)16151-5 PMID: 15135599.
- Remme J, Dadzie KY, Rolland A, Thylefors B. Ocular onchocerciasis and intensity of infection in the community. I. West African savanna. Trop Med Parasitol 1989; 40(3):340–7. PMID: 2617045.
- Boussinesq M, Gardon J, Kamgno J, Demanga N, Pion SD, Duke BO. Studies on the macrofilarial population of Onchocerca volvulus in hyper-endemic villages of the Central province of Cameroon. Ann Trop Med Parasitol 2001; 95(4):379–88. https://doi.org/10.1080/00034980120064337 PMID: 11454247.
- Abd-Elfarag G, Carter JY, Raimon S, Sebit W, Suliman A, Fodjo JNS, et al. Persons with onchocerciasis-associated epilepsy and nodding seizures have a more severe form of epilepsy with more cognitive impairment and higher levels of Onchocerca volvulus infection. Epileptic Disord 2020. <u>https://doi.org/</u> 10.1684/epd.2020.1164 PMID: 32540799.
- 85. Seidenfaden R, Fischer A, Bonow I, Ekale D, Tanya V, Renz A. Combined benefits of annual mass treatment with ivermectin and cattle zooprophylaxis on the severity of human onchocerciasis in

northern Cameroon. Trop Med Int Health 2001; 6(9):715–25. https://doi.org/10.1046/j.1365-3156. 2001.00771.x PMID: 11555439.

- Wahl G, Enyong P, Ngosso A, Schibel JM, Moyou R, Tubbesing H, et al. Onchocerca ochengi: epidemiological evidence of cross-protection against Onchocerca volvulus in man. Parasitology 1998; 116 (Pt 4):349–62. https://doi.org/10.1017/s003118209700228x PMID: 9585937.
- Colebunders R, Hendy A, van Oijen M. Nodding Syndrome in Onchocerciasis Endemic Areas. Trends Parasitol 2016; 32(8):581–3. https://doi.org/10.1016/j.pt.2016.05.013 PMID: 27289272.
- Lewis DJ. Simulium Damnosum and its relation to onchocerciasis in the Anglo-Egyptian Sudan. Bull Entomol Res 1953; 43(4):597–644. https://doi.org/10.1017/S0007485300026705
- 89. Colebunders R, Hendy A, Mokili JL, Wamala JF, Kaducu J, Kur L, et al. Nodding syndrome and epilepsy in onchocerciasis endemic regions: comparing preliminary observations from South Sudan and the Democratic Republic of the Congo with data from Uganda. BMC Res Notes 2016; 9:182. https://doi.org/10.1186/s13104-016-1993-7 PMID: 27005304; PubMed Central PMCID: PMC4802870.
- Hotterbeekx A, Dusabimana A, Mandro M, Abhafule GM, Deogratias W, Siewe Fodjo JN, et al. Urinary N-acetyltyramine-O,beta-glucuronide in Persons with Onchocerciasis-Associated Epilepsy. Pathogens. 2020; 9(3). https://doi.org/10.3390/pathogens9030191 PMID: 32151038.
- Mandro M, Siewe Fodjo JN, Dusabimana A, Mukendi D, Haesendonckx S, Lokonda R, et al. Single versus Multiple Dose Ivermectin Regimen in Onchocerciasis-Infected Persons with Epilepsy Treated with Phenobarbital: A Randomized Clinical Trial in the Democratic Republic of Congo. Pathogens. 2020; 9(3). https://doi.org/10.3390/pathogens9030205 PMID: 32164367.
- 92. Johnson TP, Tyagi R, Lee PR, Lee MH, Johnson KR, Kowalak J, et al. Nodding syndrome may be an autoimmune reaction to the parasitic worm Onchocerca volvulus. Sci Transl Med. 2017; 9(377). https://doi.org/10.1126/scitranslmed.aaf6953 PMID: 28202777; PubMed Central PMCID: PMC5434766.
- Johnson T, Nutman T, Dowell S, Nath A. Closing the Loop between Nodding Syndrome and Onchocerca Infection. Trends Parasitol 2017; 33(7):490. <u>https://doi.org/10.1016/j.pt.2017.05.002</u> PMID: 28596063.
- 94. Nanda V, Wang T, Pjanic M, Liu B, Nguyen T, Matic LP, et al. Functional regulatory mechanism of smooth muscle cell-restricted LMOD1 coronary artery disease locus. PLoS Genet 2018; 14(11): e1007755. https://doi.org/10.1371/journal.pgen.1007755 PMID: 30444878; PubMed Central PMCID: PMC6268002 (CGN) together with AR. OF is a part-time employee of CGN. JLMB, AR and EES are members of the board of directors. CGN has an invested interest in the STARNET biobank and data set. However, CGN has expressed no claims or sought any patents related to the results presented in this manuscript.
- Pollanen MS, Onzivua S, Robertson J, McKeever PM, Olawa F, Kitara DL, et al. Nodding syndrome in Uganda is a tauopathy. Acta Neuropathol 2018; 136(5):691–7. https://doi.org/10.1007/s00401-018-1909-9 PMID: 30219940; PubMed Central PMCID: PMC6208726.
- Hotterbeekx A, Lammens M, Idro R, Akun PR, Lukande R, Akena G, et al. Neuroinflammation and Not Tauopathy Is a Predominant Pathological Signature of Nodding Syndrome. J Neuropathol Exp Neurol 2019; 78(11):1049–58. <u>https://doi.org/10.1093/jnen/nlz090</u> PMID: <u>31553445</u>; PubMed Central PMCID: PMC6839030.
- 97. Levite M, Zelig D, Friedman A, Ilouz N, Eilam R, Bromberg Z, et al. Dual-Targeted Autoimmune Sword in Fatal Epilepsy: Patient's glutamate receptor AMPA GluR3B peptide autoimmune antibodies bind, induce Reactive Oxygen Species (ROS) in, and kill both human neural cells and T cells. J Autoimmun 2020:102462. https://doi.org/10.1016/j.jaut.2020.102462 PMID: 32561150.
- Levite M. Glutamate receptor antibodies in neurological diseases. J Neural Transm (Vienna) 2014; 121(8):1029–75. https://doi.org/10.1007/s00702-014-1193-3 PMID: 25081016.
- 99. Benedek G, Abed El Latif M, Miller K, Rivkin M, Ramadhan Lasu AA, Riek LP, et al. Protection or susceptibility to devastating childhood epilepsy: Nodding Syndrome associates with immunogenetic fingerprints in the HLA binding groove. PLoS Negl Trop Dis. 2020; 14(7):e0008436. https://doi.org/10. 1371/journal.pntd.0008436 PMID: 32639997; PubMed Central PMCID: PMC7371228.
- Hissette J. Memoire sur l'Onchocerca volvulus "Luecart" et ses manifestations oculaires au Congo Belge. Ann. de Méd Tropical, Brux1932.
- 101. Winkler AS, Friedrich K, Velicheti S, Dharsee J, Konig R, Nassri A, et al. MRI findings in people with epilepsy and nodding syndrome in an area endemic for onchocerciasis: an observational study. Afr Health Sci 2013; 13(2):529–40. <u>https://doi.org/10.4314/ahs.v13i2.51</u> PMID: 24235964; PubMed Central PMCID: PMC3824510.
- 102. Hotterbeekx A, Raimon S, Abd-Elfarag G, Carter JY, Sebit W, Suliman A, et al. Kumar-Singh S. Onchocerca volvulus is not detected in the cerebrospinal fluid of persons with onchocerciasis-associated epilepsy Int J Infect Dis In press 2019. https://doi.org/10.1016/j.ijid.2019.11.029 PMID: 31786246

- 103. Tai XY, Koepp M, Duncan JS, Fox N, Thompson P, Baxendale S, et al. Hyperphosphorylated tau in patients with refractory epilepsy correlates with cognitive decline: a study of temporal lobe resections. Brain 2016; 139(Pt 9):2441–55. https://doi.org/10.1093/brain/aww187 PMID: 27497924; PubMed Central PMCID: PMC5926008.
- 104. Wu JW, Hussaini SA, Bastille IM, Rodriguez GA, Mrejeru A, Rilett K, et al. Neuronal activity enhances tau propagation and tau pathology in vivo. Nat Neurosci 2016; 19(8):1085–92. <u>https://doi.org/10.1038/</u> nn.4328 PMID: 27322420; PubMed Central PMCID: PMC4961585.
- 105. Santangelo R, Cecchetti G, Bernasconi MP, Cardamone R, Barbieri A, Pinto P, et al. Cerebrospinal Fluid Amyloid-beta 42, Total Tau and Phosphorylated Tau are Low in Patients with Normal Pressure Hydrocephalus: Analogies and Differences with Alzheimer's Disease. J Alzheimers Dis 2017; 60 (1):183–200. https://doi.org/10.3233/JAD-170186 PMID: 28826180.
- 106. Colebunders R, Abd-Elfarag G, Carter JY, Olore PC, Puok K, Menon S, et al. Clinical characteristics of onchocerciasis-associated epilepsy in villages in Maridi County, Republic of South Sudan Seizure 2018; 62:108–15. https://doi.org/10.1016/j.seizure.2018.10.004 PMID: 30340162.
- 107. Abd-Elfarag G, Logora MY, Carter JY, Ojok M, Songok J, Menon S, et al. The effect of bi-annual community-directed treatment with ivermectin on the incidence of epilepsy in onchocerciasis endemic villages in South Sudan: a study protocol. Infect Dis Poverty 2018; 7(1):112. https://doi.org/10.1186/ s40249-018-0496-2 PMID: 30424817; PubMed Central PMCID: PMC6234538.
- 108. Makepeace BL, Tanya VN. 25 Years of the Onchocerca ochengi Model. Trends Parasitol 2016; 32 (12):966–78. https://doi.org/10.1016/j.pt.2016.08.013 PMID: 27665524.
- 109. Otranto D, Giannelli A, Scotty Trumble N, Chavkin M, Kennard G, Latrofa MS, et al. Clinical case presentation and a review of the literature of canine onchocercosis by Onchocerca lupi in the United States. Parasit Vectors 2015; 8:89. https://doi.org/10.1186/s13071-015-0699-3 PMID: 25884672; PubMed Central PMCID: PMC4346121.
- 110. Twum-Danso NA. Loa loa encephalopathy temporally related to ivermectin administration reported from onchocerciasis mass treatment programs from 1989 to 2001: implications for the future. Filaria J 2003; 2 Suppl 1:S7. https://doi.org/10.1186/1475-2883-2-S1-S7 PMID: 14975064; PubMed Central PMCID: PMC2147656.
- 111. Sakti H, Nokes C, Hertanto WS, Hendratno S, Hall A, Bundy DA, et al. Evidence for an association between hookworm infection and cognitive function in Indonesian school children. Trop Med Int Health 1999; 4(5):322–34. https://doi.org/10.1046/j.1365-3156.1999.00410.x PMID: 10402967.
- 112. Chao PC, Chien WC, Chung CH, Chu CW, Yeh CB, Chang HA, et al. Pinworm infections associated with risk of psychiatric disorders-A nationwide cohort study in Taiwan: Pinworm infections and psychiatric disorders. Compr Psychiatry 2019; 93:14–9. https://doi.org/10.1016/j.comppsych.2019.02.002 PMID: 31280142.
- 113. Brattig NW. Pathogenesis and host responses in human onchocerciasis: impact of Onchocerca filariae and Wolbachia endobacteria. Microbes Infect 2004; 6(1):113–28. <u>https://doi.org/10.1016/j.micinf.</u> 2003.11.003 PMID: 14738900.
- 114. Budden FH. The incidence of microfilariae in the eye and of ocular lesions in relation to the age and sex of persons living in communities where onchocerciasis is endemic. Trans R Soc Trop Med Hyg 1963; 57:71–5. https://doi.org/10.1016/0035-9203(63)90014-2 PMID: 14016593.
- Budden FH. Ocular Lesions of Onchocerciasis. Br J Ophthalmol. 1962; 46(1):1–11. https://doi.org/10. 1136/bjo.46.1.1 PMID: 18170747; PubMed Central PMCID: PMC505726.
- 116. Zhou Y, Dziak E, Unnasch TR, Opas M. Major retinal cell components recognized by onchocerciasis sera are associated with the cell surface and nucleoli. Invest Ophthalmol Vis Sci 1994; 35(3):1089–99. PMID: 7510272.
- 117. McKechnie NM, Gurr W, Braun G. Immunization with the cross-reactive antigens Ov39 from Onchocerca volvulus and hr44 from human retinal tissue induces ocular pathology and activates retinal microglia. J Infect Dis 1997; 176(5):1334–43. https://doi.org/10.1086/514130 PMID: 9359736.
- 118. Colebunders R, Mandro M, Mukendi D, Dolo H, Suykerbuyk P, Van Oijen M. Ivermectin Treatment in Patients With Onchocerciasis-Associated Epilepsy: Protocol of a Randomized Clinical Trial. JMIR Res Protoc 2017; 6(8):e137. <u>https://doi.org/10.2196/resprot.7186</u> PMID: <u>28855148</u>; PubMed Central PMCID: PMC5597797.
- 119. Idro R, Anguzu R, Ogwang R, Akun P, Abbo C, Mwaka AD, et al. Doxycycline for the treatment of nodding syndrome (DONS); the study protocol of a phase II randomised controlled trial. BMC Neurol 2019; 19(1):35. https://doi.org/10.1186/s12883-019-1256-z PMID: 30841858; PubMed Central PMCID: PMC6402111.
- 120. Ioannidis JP. Exposure-wide epidemiology: revisiting Bradford Hill. Stat Med 2016; 35(11):1749–62. https://doi.org/10.1002/sim.6825 PMID: 26646432.

- 121. Lucas RM, McMichael AJ. Association or causation: evaluating links between "environment and disease". Bull World Health Organ 2005; 83(10):792–5. https://doi.org//S0042-96862005001000017 PMID: 16283057; PubMed Central PMCID: PMC2626424.
- 122. Colebunders R, Basanez MG, Siling K, Post RJ, Rotsaert A, Mmbando B, et al. From river blindness control to elimination: bridge over troubled water. Infect Dis Poverty 2018; 7(1):21. <u>https://doi.org/10.1186/s40249-018-0406-7 PMID: 29587844</u>; PubMed Central PMCID: PMC5872540.
- 123. Sejvar JJ, Kakooza AM, Foltz JL, Makumbi I, Atai-Omoruto AD, Malimbo M, et al. Clinical, neurological, and electrophysiological features of nodding syndrome in Kitgum, Uganda: an observational case series. Lancet Neurol 2013; 12(2):166–74. https://doi.org/10.1016/S1474-4422(12)70321-6 PMID: 23305742.
- 124. Remo ML. Novel evidence to motivate onchocerciasis elimination Lancet Infectious Diseases. 2020 Jun 26;S1473-3099(20)30284-X.